Author’s response to reviews

Title: The influence of the rs1137101 genotypes of leptin receptor gene on the demographic and metabolic profile of normal Saudi females and those suffering from polycystic ovarian syndrome

Authors:

Maha Daghestani (mdaghestani@ksu.edu.sa)
Mazen Daghestani (mhdaghestani@uqu.edu.sa)
Mamoon Daghistani (mdaghist@hotmail.com)
Geir Bjørklund (bjorklund@conem.org)
Salvatore Chirumbolo (salvatore.chirumbolo@univr.it)
Arjumand Warsy (aswarsy@gmail.com; aswarsy@ksu.edu.sa)

Version: 2 Date: 19 Sep 2017

Author’s response to reviews:

To: "BMC Women's Health - Editorial Office"

Subject: BMWH-D-17-00231R1

Dear Sir

Thank you for your letter regarding my paper mentioned above. I have revised the paper accordingly. The changes made in the paper are as follows:

1- A Conclusion section is incorporated after the Discussion Section.

2- "Declaration" Section is incorporated at the end of the manuscript. It contains list of abbreviations, ethical approval and consent to participate, consent for publication, availability of data and materials, competing interest, funding, authors contributions and acknowledgement sections.

I do hope I have fulfilled all requirement of the Journal and in the present form the paper fulfills all requirements of 'BMC Women's Health'.

I look forward to hearing from you,

Best regards

Arjumand Warsy